%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% AGUtmpl.tex: this template file is for articles formatted with LaTeX2e,
% Modified March 7, 2008, 3:30pm
%
% This template includes with commands and instructions
% given in the order necessary to produce a final output that will
% satisfy AGU requirements.
%
% PLEASE DO NOT USE YOUR OWN MACROS
%
% For more information on using the AGUTeX macro package,
% see agudocs.tex or agudocs.pdf
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% All questions should be e-mailed to author.help@agu.org.
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Step 1: set the \documentclass
%
% The three options for article format are: two-column (default),
% draft, for initial article submission; and galley for narrow
% single columns.
%
% PLEASE USE THE DRAFT OPTION TO SUBMIT YOUR PAPERS
% The draft option produces double spaced output
%
% Choose the journal abbreviation for the journal you are
% submitting to:

% jgrga JOURNAL OF GEOPHYSICAL RESEARCH
% gbc   GLOBAL BIOCHEMICAL CYCLES
% grl   GEOPHYSICAL RESEARCH LETTERS
% pal   PALEOCEANOGRAPHY
% ras   RADIO SCIENCE
% rog   REVIEWS OF GEOPHYSICS
% tec   TECTONICS
% wrr   WATER RESOURCES RESEARCH
% gc    GEOCHEMISTRY, GEOPHYSICS, GEOSYSTEMS

% (If you are submitting to a journal other than jgrga,
% substitute the initials of the journal for "jgrga" below)

\documentclass[draft,gbc]{agutex}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%% optional article formats author might want to use

% To produce a galley version:
% \documentclass[galley,jgrga]{AGUTeX}

% To produce a two columned version:
% \documentclass[jgrga]{AGUTeX}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% OPTIONAL:
% To print your article using PostScript fonts, uncomment this:
% \usepackage{agu-ps}
% You many need to edit the top of agu-ps to use the names of the PS
% fonts on your system.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% OPTIONAL:
% To Create numbered lines:

% If you don't already have lineno.sty, you can download it from
% http://www.ctan.org/tex-archive/macros/latex/contrib/ednotes/
% (or google lineno.sty ctan), available at TeX Archive Network (CTAN).
% Take care that you always use the latest version.

% To activate the commands, uncomment \usepackage{lineno}
% and \linenumbers*[1]command, below:

%\usepackage{lineno}

%\linenumbers*[1]


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Figures and Tables
%

% When submitting articles through the GEMS system:
% COMMENT OUT ANY COMMANDS THAT INCLUDE GRAPHICS.
% (See FIGURES section near the end of the file)


%  Figures and Tables should be placed at the end of the article,
%  after the references.
%
%  Uncomment the following line to include .eps files:
%  \usepackage[dvips]{graphicx}
  \usepackage{lscape,pstricks}
%
%    Also uncomment the following command to allow illustrations to print
%    when using Draft:
%\setkeys{Gin}{draft=false}
%
% Substitute one of the following for [dvips] above
% if you are using a different driver program and want to
% proof the your illustrations on your machine:
%
% [xdvi], [dvipdf], [dvipsone], [dviwindo], [emtex], [dviwin],
% [pctexps],  [pctexwin],  [pctexhp],  [pctex32], [truetex], [tcidvi],
% [oztex], [textures]
%
% See how to enter figures and tables at the end of the article, after
% references.
%
%% ------------------------------------------------------------------------ %%
%
%  ENTER PREAMBLE
%
%% ------------------------------------------------------------------------ %%

% Author names in capital letters:
\authorrunninghead{FR\"{o}LICHER ET AL.}

% Shorter version of title entered in capital letters:
\titlerunninghead{VARIABILITY AND TRENDS IN OCEANIC OXYGEN}

% Author address will appear at end of article, please repeat
% this command for each author:

%\authoraddr{T. L. Fr\"{o}licher,
%Climate and Environmental Physics, University of Bern,
%Sidlerstrasse 5, CH-3012 Bern, Switzerland.
%(froelicher@climate.unibe.ch)}


\begin{document}
\begin{thebibliography}{}
% \bibitem
% Text
% \end{thebibliography}


%% ------------------------------------------------------------------------ %%
%
%  TITLE
%
%% ------------------------------------------------------------------------ %%

\title{Text S1}
%\title{Natural variability and anthropogenic trends in oceanic oxygen in a coupled carbon cycle-climate model ensemble - supplementary material}
%
% e.g., \title{Terrestrial Ring Current:
% Origin, Formation and Decay $\alpha\beta\Gamma\Delta$}
% You may use \\ to break the title over several lines.

%% ------------------------------------------------------------------------ %%
%
%  AUTHORS AND AFFILIATIONS - 2 methods
%
%% ------------------------------------------------------------------------ %%

% Method 1
% For three or fewer author/affiliation blocks, use \author{} and \affil{}
%\date{intend to submit to GLOBAL BIOCHEMICAL CYCLES: \today}
% \authors{T. L. Fr\"{o}licher, \altaffilmark{1}
%  F. Joos, \altaffilmark{1,2}
%  G. -K. Plattner, \altaffilmark{3}
%  M. Steinacher, \altaffilmark{1} and
%  S. C. Doney\altaffilmark{4}}
%\textbf{intend to submit to Global Biogeochemical Cycles} %\today
% \altaffiltext{1}
% {Climate and Environmental Physics, University of Bern, Bern, Switzerland}
% \altaffiltext{2}
% {Oeschger Centre for Climate Change Research, University of Bern, Bern, Switzerland}
% \altaffiltext{3}
% {Environmental Physics, Institute of Biogeochemistry and Pollutant Dynamics  ETH Z\"{u}rich, Z\"{u}rich, Switzerland}
% \altaffiltext{4}
% {Marine Chemistry and Geochemistry, Woods Hole Oceanographic Institution, Woods Hole, USA}


%\author{T. L. Fr\"{o}licher,$^1$ F. Joos,$^{1,2}$ G. -K. Plattner,$^3$  M. Steinacher,$^1$ and S. C. Doney$^4$}

%\affil{$^1$Climate and Environmental Physics, University of Bern, Bern, Switzerland.}
%\affil{$^2$Oeschger Centre for Climate Change Research, University of Bern, Bern, Switzerland.}
%\affil{$^3$Environmental Physics, Institute of Biogeochemistry and Pollutant Dynamics, ETH Z\"{u}rich, Z\"{u}rich, Switzerland.}
%\affil{$^4$Marine Chemistry and Geochemistry, Woods Hole Oceanographic Institution, Woods Hole, USA.}

% ---------------
% Method 2
% For more than three author/affiliation blocks,
% use \author{\altaffilmark{}} and \altaffiltext{}
% \altaffilmark will produce footnote;
% matching altaffiltext will appear at bottom of page.
% May use \\ to start a new line.

% \authors{R. C. Bales, \altaffilmark{1}
% E. Mosley-Thompson, \altaffilmark{2}
% R. Williams, \altaffilmark{3}
% and J. R. McConnell\altaffilmark{4}}

% \altaffiltext{1}
% {Department of Hydrology and Water Resources, University of Arizona,
% Tucson, Arizona, USA.}
%
% \altaffiltext{2}{Department of Geography, Ohio State University,
% Columbus, Ohio, USA.}
%
% \altaffiltext{3}{Department of Space Sciences, University of Michigan,
% Ann Arbor, Michigan, USA.}
%
% \altaffiltext{4}{Desert Research Institute, Division of Hydrologic Sciences,
% Reno, Nevada, USA.}

%% ------------------------------------------------------------------------ %%
%
%  ABSTRACT
%
%% ------------------------------------------------------------------------ %%

% >> Do NOT include any \begin...\end commands within
% >> the body of the abstract.



%% ------------------------------------------------------------------------ %%
%
%  BEGIN ARTICLE
%
%% ------------------------------------------------------------------------ %%

% The body of the article must start with a \begin{article} command
%
% \end{article} must follow the references section, before the figures
%  and tables.

\begin{article}

%% ------------------------------------------------------------------------ %%
%
%  TEXT
%
%% ------------------------------------------------------------------------ %%
\section{Biogeochemical Model}
The ocean biogeochemical model includes in simplified form the main physico-chemical and biological processes governing the cycling of the tracers dissolved inorganic carbon, alkalinity, phosphate, iron and oxygen and the export and remineralization fluxes of dissolved and particulate organic carbon and calcium carbonate (CaCO$_3$). In the ocean model, the biological source-sink term has been changed from a nutrient restoring formulation as used in OCMIP-2 \citep{Najjar:07} to a prognostic formulation. Biological productivity is computed prognostically from temperature, surface solar irradiance, mixed layer depth, and macro- and micronutrients. Following the OCMIP-2 protocols, total biological productivity is partitioned 1/3 into sinking particulate organic matter (POC), and 2/3 into the formation of dissolved or suspended organic matter, much of which is remineralized within the model euphotic zone. Total productivity thus contains both new (upwelling or upward mixing of nutrients from the thermocline) and regenerated (recycling of organic matter within the surface ocean) production. While not strictly equivalent to primary production as measured by $^{14}$C methods, model primary production is a reasonable proxy for the time and space variability of observed primary production, but has a lower magnitude. Since the model doesn't explicitly include sediments, any organic matter reaching the bottom of the ocean is assumed to be remineralized in the lowest layer of the model with a first-order rate. The marine iron cycle is parametrized following \cite{Doney:06} including atmospheric dust deposition/iron dissolution, biological uptake, vertical particle transport and scavenging.\\
The modeled atmospheric oxygen mass balance equation is given by
%\begin{linenomath*}
\begin{eqnarray}
\eqnum{S1}
  \mathrm{\frac{\partial[O_2]_{atm}}{\partial t}} + \mathrm{\xi} =  - \mathrm{\Phi_{foss}} + \mathrm{\Phi_{oa}} + \mathrm{\Phi_{ta}}.
\label{ox_flux}
\end{eqnarray}
%\end{linenomath*}
Atmospheric O$_2$, \lbrack{O$_2$}\rbrack$_{atm}$, is advected and mixed as a dry-air mixing ratio using a semi-Lagrangian advection scheme, where $\xi$ is the atmospheric transport operator for oxygen. $\Phi$ represent the source-sink terms due to O$_2$ consumption by fossil fuel burning ($\Phi$$_{foss}$), the net exchange between the ocean and the atmosphere ($\Phi$$_{oa}$), and the net exchange between the terrestrial biosphere and the atmosphere ($\Phi$$_{ta}$). \\
Oxygen concentrations in the ocean interior, \lbrack{O$_2$}\rbrack, are governed by
%\begin{linenomath*}
\begin{eqnarray}
\eqnum{S2} \mathrm{\frac{\partial[O_2]}{\partial t}}+ \mathrm{L} =
\mathrm{\Phi_{b}} + \mathrm{\Phi_{ao}},
 \label{ox_conc}
\end{eqnarray}
%\end{linenomath*}
where L is the 3-D transport operator for oxygen, which represents effects due to advection, diffusion, and convection. $\Phi_{b}$ is the biological source-sink term for oxygen and $\Phi_{ao}$ is the source-sink term as a result of air-sea gas exchange. The production and remineralization of organic phosphorus produces and consumes O$_2$ in a constant stoichiometric O$_2$:P ratio of -170 mol O$_2$ per mol P \citep{Anderson:94}. Denitrification in oxygen minimum zones is dealt with implicitly in the model by halting oxygen consumption below an oxygen threshold ([O$_2$]$_{crit}$ = 4 $\mu$mol kg$^{-1}$). The ocean will then be a net source of oxygen to the atmosphere because oxygen production above the compensation depth will be greater than oxygen consumption below the compensation depth.\\
Air-sea fluxes of O$_2$ are computed according to OCMIP-2, except that surface wind speed, atmospheric pressure, and ice fraction are from the atmosphere and sea-ice components. The source-sink term for the surface layer is:
%\begin{linenomath*}
\begin{eqnarray}
\eqnum{S3} \mathrm{\Phi_{ao}} =
\mathrm{\frac{k_{w}}{dz_1}}*\mathrm{([O_2]^{surf} -
\frac{P^{0}}{P}*[O_2]^{0}_{sat})}, \label{ox_ocmip}
\end{eqnarray}
%\end{linenomath*}
where [O$_2$]$^{surf}$ is the oxygen concentration in the top layer of the ocean (dz$_1$ = 12 m) and [O${_2}$]$^{0}$$_{sat}$ is the oxygen saturation concentration at surface pressure (P$^{0}$ = 1 atm) in mol m$^{-3}$ for a given temperature and salinity \citep{Garcia:92}. P is the total atmospheric pressure and the windspeed-dependent gas exchange coefficient k$_{w}$ across the air-sea interface is adopted from \cite{Wanninkhof:92}:
%\begin{linenomath*}
\begin{eqnarray}
\eqnum{S4} \mathrm{k_{w}} =
\mathrm{(1-f_{ice})*0.31*U_{10}^{2}*(S_{c}/660)^{-1/2}},
\label{ox_wan}
\end{eqnarray}
%\end{linenomath*}
%\end{linenomath*}
where k$_w$ has units of cm hr$^{-1}$ and U$_{10}$ is the wind speed in ms$^{-1}$ interpolated to a height of 10 m above the sea level. f$_{ice}$ is the fractional sea-ice coverage and the Schmidt number S$_c$ for oxygen is computed  using the formulation proposed by \cite {Keeling:02}.
\section{Climatological Mean State of AOU: Data versus Model Results}
\label{climatological_mean_state}
The modeled large-scale pattern of AOU compares reasonably well with observations from the World Ocean Atlas 2005 \citep{Garcia:06} (Figure S\ref{AOU_mean_state}). However, the AOU maximum in the equatorial Atlantic is slightly overestimated because biological production (as defined in section 2 in the supplementary material) is too high compared with observations as a consequence of too intense upwelling \citep{Schneider:08}. An obvious difference between model and observations is the much shallower penetration of North Atlantic Deep Water (NADW) in the model indicated by high concentrations of AOU in the deep North Atlantic. The model underestimates AOU concentrations (overestimates O$_2$) by up to 100 $\mu$mol kg$^{-1}$ in the North Pacific. This is caused by too little nutrient inputs into the upper, sunlit surface ocean, overly strong iron limitation and consequently too low export of organic matter in the North Pacific, and problems with upwelling parametrization \citep{Schneider:08,Doney:06}. Excessive iron limitation occurs mainly in the subtropical Pacific and is linked to overly strong iron scavenging.
\section{Oxygen Drift and Detrending Procedure}
For this study, all runs were performed on an IBM SP4 and on an IBM P5 at the Swiss National Supercomputing Centre. Results for a benchmark simulation over the historical period agree with the results presented by \cite{Fung:05}. The control simulation exhibits a drift in dissolved oxygen concentrations and other tracers related to the long time-scale of deep ocean overturning. The total oxygen inventory is increasing by about 0.9 Pmol over the simulated period (Figure 2a, main text). The increase in oxygen occurs at intermediate depth (1500 m) in the Pacific and oxygen is decreasing at deeper depth, mainly in the Atlantic Ocean. The total sea-air O$_2$ flux is increasing by about 0.01 Pmol over the simulated period with a permanent outgassing of 0.03 Pmol yr$^{-1}$. Thus, we apply a grid-point based correction to remove model drift. This is done by splining monthly-mean values of the control simulation at each grid point and for each calender month individually with a spline that eliminates high-frequency variations (nominal cut-off period of 200 years for dissolved oxygen, air-sea oxygen flux, phosphate (PO$_4$) and dissolved inorganic carbon (DIC), and of 1000 years for temperature and salinity). These splines, representing low-frequency trends, are then subtracted from the results of the transient simulations. The sea-air O$_2$ flux offset of 0.03 Pmol yr$^{-1}$ is subtracted for analysis in section 3.1. One reason for the slight trend in dissolved O$_2$ is drift in the coupled physical simulation resulting in surface ocean freshening and a corresponding increase in deep-water salinity due to problems in the representation of sea-ice. Additionally, the ocean oxygen is not in balance (i.e not conserved) in the ocean biogeochemistry model (as defined in OCMIP-2 BGC models) because of denitrification.
\section{Definitions of Different Internal Oscillation Modes}
We calculated the different circulation indices from a 480 year control run. The NAO index is calculated as the amplitude of the pattern defined by the leading empirical orthogonal function (EOF) of winter (DJFM) monthly mean surface layer pressure anomalies between 20$^\circ$N and 70$^\circ$N and between 90$^\circ$W and 40$^\circ$E \citep{Hurrell:95}. The simulated explained variances of the NAO is 51 $\%$. The PDO index is computed from the first orthogonal function of SST over the North Pacific north of 20$^\circ$N \citep{Deser:04}. The explained variance of the PDO index is 34 $\%$. The performance of the model in representing the different climate indices is described in detail by \cite{Meehl:98}.
\section{Data Sources and Interpolation Method}
Data used in section 3.3 of the main text for illustrative model-data comparisons have been sampled along 152$^\circ$W in the eastern subtropical/subarctic North Pacific in the years 1980, 1984, 1991 and 1997 \citep{Deutsch:06,Emerson:04,Emerson:01}. In the Atlantic, data along 20$^\circ$W starting south of Iceland are used. The Atlantic sections were sampled in the years 1988 \citep{Tsuchiya:92}, 1993 \citep{Doney:98}, 1998 \citep{Smythe-Wright:99} and 2003 ($\it{http://ushydro.ucsd.edu}$), and are discussed in \cite{Johnson:07}.\\
Observed AOU and O$_2$ changes in the North Atlantic and Pacific have been interpolated onto a grid with a horizontal resolution of 1$^\circ$ $\times$ 1$^\circ$ and potential density as vertical coordinate. Potential density is referenced to 1000 dbar ($\sigma _1$) and 0 dbar ($\sigma  $), respectively. A Gaussian interpolation function with a 2$^\circ$ mapping scale for Gaussian weights is used. Modeled tracer concentrations are converted from volume normalized units to conventional mass normalized units using a constant conversion factor of 1024.5 kg m$^{-3}$.

%%% End of body of article:

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% Optional Appendix goes here
%
%%%%%%%%%%%%%%%%%
% Geophysical Research Letters only allows an appendix without a letter.
%% You can get this result with
%  \section*{Appendix}
%  or
%  \section*{Appendix: Title}
%%%%%%%%%%%%%%%%%
%
% \appendix resets counters and redefines section heads
% but doesn't print anything.
% After typing  \appendix
%
% \section{Here Is Appendix Title}
% will print
% Appendix A: Here Is Appendix Title
%
% \section*{Appendix}
% will print
% Appendix
%
% \section*{Appendix: Here Is Appendix Title}
% will print
% Appendix: Here Is Appendix Title
%
% For only 1 appendix \appendix \section{Appendix} is preferred.
% which will print
% Appendix A

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Optional Glossary or Notation section, goes here
%
%%%%%%%%%%%%%%
% Glossary only allowed in Reviews of Geophysics
% \section*{Glossary}
% \paragraph{Term}
% Term Definition here
%
%%%%%%%%%%%%%%
% Notation -- End each entry with a period.
% \begin{notation}
% Term & definition.\\
% Second Term & second definition.
% \end{notation}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%  ACKNOWLEDGMENTS



%% ------------------------------------------------------------------------ %%
%
%  REFERENCE LIST AND TEXT CITATIONS
%
% Either type in your references using
% 
%
% Or,
%
% If you use BiBTeX for your References, please produce your .bbl
% file and copy the contents into your paper here.
%
% Follow these steps:
% 1. Run LaTeX on your LaTeX file.
%
% 2. Run BiBTeX on your LaTeX file.
%
% 3. Open the new .bbl file containing the reference list and
%   copy all the contents into your LaTeX file here.
%
% 4. Comment out the old \bibliographystyle and \bibliography commands.
%
% 5. Run LaTeX on your new file before submitting.
%
% AGU does not want a .bib or a .bbl file, but asks that you
% copy in the contents of your .bbl file here.

%\begin{thebibliography}{}
%\bibitem{jskilby}J. S. Kilby,
%``Invention of the Integrated Circuit,'' {\it IEEE Trans. Electron Devices,}
%{\bf ED-23,} 648 (2008).
%\end{thebibliography}

%% ------------------------------------------------------------------------ %%
%
%  END ARTICLE
%
%% ------------------------------------------------------------------------ %%




%% Enter Figures and Tables at here:

% When submitting articles through the GEMS system:
% COMMENT OUT ANY COMMANDS THAT INCLUDE GRAPHICS.

% Figure captions go below this illustration; Table captions above tables

% ONE-COLUMN figure/table, including eps graphics
%
% \begin{figure}
% \noindent\includegraphics[width=20pc]{samplefigure.eps}
% \caption{Caption text here}
% \end{figure}
% \end{document}
%
% \begin{table}
% \caption{}
% \end{table}
%
% ---------------
% TWO-COLUMN figure/table
%
% \begin{figure*}
% \noindent\includegraphics[width=39pc]{samplefigure.eps}
% \caption{Caption text here}
% \end{figure*}
%
% \begin{table*}
% \caption{Caption text here}
% \end{table*}
%
% see below for how to make landscape figures or tables
% Table 3
%\begin{figure*}
%\begin{center}
%\noindent\includegraphics[width=35pc]{/home/froelicher/paper/oxygen/fig/summary_cruises.eps}
% \caption{Summary of observed long-term changes in the ocean's oxygen content reported in literature and partly used in this study (shown by the red labels). The color-scale indicate the maximum AOU changes in the water column over the years. The black lines show only the locations of the repeated cruises. The year of sampling is indicated by the black labels.}
% \label{summary_of_studies}
%\end{center}
%\end{figure*}
\setfigurenum{S1}
\begin{figure*}
\begin{center}
%\noindent\includegraphics[width=37pc]{/home/froelicher/paper/oxygen/tex/gbc/fig/sup_fig1/aou_mean_state.eps}
 \caption{Observed (top) and modeled (bottom) zonal mean of AOU for the Atlantic Ocean (left) and the Indo-Pacific Ocean (right). Observed distributions are from the World Ocean Atlas 2005 \citep{Garcia:06}. The model ensemble mean is averaged from 1990 to 1999. Contours are every 1 $\mu$mol kg$^{-1}$ and represent one standard deviation among the six ensemble members for the zonal and decadal mean.}
 \label{AOU_mean_state}
\end{center}
\end{figure*}



%\bibliographystyle{/home/froelicher/paper/bib/agufull04}
%\bibliographystyle{agufull}
%\bibliography{/home/froelicher/paper/bib/bibliography_froe.bib}
\begin{thebibliography}{19}
\providecommand{\natexlab}[1]{#1}
\expandafter\ifx\csname urlstyle\endcsname\relax
  \providecommand{\doi}[1]{doi:\discretionary{}{}{}#1}\else
  \providecommand{\doi}{doi:\discretionary{}{}{}\begingroup
  \urlstyle{rm}\Url}\fi

\bibitem[{\textit{Anderson and Sarmiento}(1994)}]{Anderson:94}
Anderson, L.~A., and J.~L. Sarmiento (1994), Redfield ratios of
  remineralization determined by nutrient data analysis, \textit{Global
  Biogeochem. Cycles}, \textit{8}, 65--80.

\bibitem[{\textit{Deser et~al.}(2004)\textit{Deser, Phillips, and
  Hurrell}}]{Deser:04}
Deser, C., A.~S. Phillips, and J.~W. Hurrell (2004), Pacific interdecadal
  climate variability: linkages between the tropics and the {N}orth {P}acific
  during boreal winter since 1900, \textit{J. Climate}, \textit{17(16)},
  3109--3124.

\bibitem[{\textit{Deutsch et~al.}(2006)\textit{Deutsch, Emerson, and
  Thompson}}]{Deutsch:06}
Deutsch, C., S.~Emerson, and L.~Thompson (2006), Physical-biological
  interactions in {N}orth {P}acific oxygen variability, \textit{J. Geophys.
  Res.}, \textit{111}, \doi{10.1029/2005JC003179}.

\bibitem[{\textit{Doney et~al.}(1998)\textit{Doney, Bullister, and
  Wanninkhof}}]{Doney:98}
Doney, S.~C., J.~L. Bullister, and R.~Wanninkhof (1998), Climate variability in
  upper ocean ventilation rates diagnosed using chlorofluorocarbons,
  \textit{Geophys. Res. Lett.}, \textit{25}, 1399--1402.

\bibitem[{\textit{Doney et~al.}(2006)\textit{Doney, Lindsay, Fung, and
  John}}]{Doney:06}
Doney, S.~C., K.~Lindsay, I.~Fung, and J.~John (2006), Natural variability in a
  stable, 1000-yr global coupled climate-carbon cycle simulation, \textit{J.
  Climate}, \textit{19}, 3033--3054.

\bibitem[{\textit{Emerson et~al.}(2001)\textit{Emerson, Mecking, and
  Abell}}]{Emerson:01}
Emerson, S., S.~Mecking, and J.~Abell (2001), The biological pump in the
  subtropical {N}orth {P}acific ocean: {N}utrient sources, redfield ratios and
  recent changes, \textit{Global Biogeochem. Cycles}, \textit{15}, 535--554.

\bibitem[{\textit{Emerson et~al.}(2004)\textit{Emerson, Watanabe, Ono, and
  Mecking}}]{Emerson:04}
Emerson, S., Y.~W. Watanabe, T.~Ono, and S.~Mecking (2004), Temporal trends in
  apparent oxygen utilization in the upper pycnocline of the {N}orth {P}acific:
  1980-2000, \textit{J. Oceanogr.}, \textit{60}, 139--147.

\bibitem[{\textit{Fung et~al.}(2005)\textit{Fung, Doney, Lindsay, and
  John}}]{Fung:05}
Fung, I.~Y., S.~C. Doney, K.~Lindsay, and J.~John (2005), Evolution of carbon
  sinks in a changing climate, \textit{Proc. Natl. Acad. Sci.},
  \textit{102(32)}, 11 201--11 206, \doi{10.1073/pnas.0504949102}.

\bibitem[{\textit{Garcia and Gordon}(1992)}]{Garcia:92}
Garcia, H., and L.~I. Gordon (1992), Oxygen solubility in sea water - better
  fitting equations, \textit{Limnol. Oceanogr.}, \textit{37}, 1307--1312.

\bibitem[{\textit{Garcia et~al.}(2006)\textit{Garcia, Locarnini, Boyer, and
  Antonov}}]{Garcia:06}
Garcia, H.~E., R.~A. Locarnini, T.~P. Boyer, and J.~I. Antonov (2006), World
  {O}cean {A}tlas 2005, {V}olume 3: {D}issolved {O}xygen, {A}pparent {O}xygen
  {U}tilization, and {O}xygen {S}aturation, \textit{Tech. rep.}, Ed. NOAA Atlas
  NESDIS 63, U.S. Government Printing Office, Silver Spring, MD.

\bibitem[{\textit{Hurrell}(1995)}]{Hurrell:95}
Hurrell, J.~W. (1995), Decadal trends in the north atlantic oscillation:
  {R}egional temperatures and precipitation, \textit{Science},
  \textit{269}(5224), 676 -- 679, \doi{10.1126/science.269.5224.676}.

\bibitem[{\textit{Johnson and Gruber}(2007)}]{Johnson:07}
Johnson, G.~C., and N.~Gruber (2007), Decadal water mass variations along
  20$^\circ${W} in the {N}ortheastern {A}tlantic {O}cean, \textit{Prog.
  Oceanogr.}, \textit{73}, 277--295, \doi{10.1016/j.pocean.2006.03.022}.

\bibitem[{\textit{Keeling et~al.}(1998)\textit{Keeling, Stephens, Najjar,
  Doney, Archer, and Heimann}}]{Keeling:02}
Keeling, R.~F., B.~B. Stephens, R.~G. Najjar, S.~C. Doney, D.~Archer, and
  M.~Heimann (1998), Seasonal variations in the atmospheric {O}$_2$/{N}$_2$
  ratio in relation to the kinetics of air-sea gas exchange, \textit{Global
  Biogeochem. Cycles}, \textit{12}, 141--164.

\bibitem[{\textit{Meehl and Arblaster}(1998)}]{Meehl:98}
Meehl, G.~A., and J.~M. Arblaster (1998), The {A}sian-{A}ustralian {M}onsoon
  and {E}l {N}ino-{S}outhern {O}scillation in the {NCAR} {C}limate {S}ystem
  {M}odel, \textit{J. Climate}, \textit{11}, 1356--1385.

\bibitem[{\textit{Najjar et~al.}(2007)\textit{Najjar, Jin, Louanchi, Aumont,
  Caldeira, Doney, Dutay, Follows, Gruber, Joos, Lindsay, Maier-Reimer, Matear,
  Matsumoto, Monfray, Mouchet, Orr, Plattner, Sarmiento, Schlitzer, Slater,
  Weirig, Yamanaka, and Yool}}]{Najjar:07}
Najjar, R.~G., X.~Jin, F.~Louanchi, O.~Aumont, K.~Caldeira, S.~C. Doney, J.~C.
  Dutay, M.~Follows, N.~Gruber, F.~Joos, K.~Lindsay, E.~Maier-Reimer, R.~J.
  Matear, K.~Matsumoto, P.~Monfray, A.~Mouchet, J.~C. Orr, G.~K. Plattner,
  J.~L. Sarmiento, R.~Schlitzer, R.~D. Slater, M.~F. Weirig, Y.~Yamanaka, and
  A.~Yool (2007), Impact of circulation on export production, dissolved organic
  matter, and dissolved oxygen in the ocean: {R}esults from {P}hase {II} of the
  {O}cean {C}arbon-{C}ycle {M}odel {I}ntercomparison {P}roject ({OCMIP}-2),
  \textit{Global Biogeochem. Cycles}, \textit{21}(GB3007),
  \doi{10.1029/2006GB002857}.

\bibitem[{\textit{Schneider et~al.}(2008)\textit{Schneider, Bopp, Gehlen,
  Segschneider, Fr{\"o}licher, Cadule, Friedlingstein, Doney, Behrenfeld, and
  Joos}}]{Schneider:08}
Schneider, B., L.~Bopp, M.~Gehlen, J.~Segschneider, T.~L. Fr{\"o}licher,
  P.~Cadule, P.~Friedlingstein, S.~C. Doney, M.~J. Behrenfeld, and F.~Joos
  (2008), Climate-induced interannual variability of marine primary and export
  production in three global coupled climate carbon cycle models,
  \textit{Biogeosciences}, \textit{5}(2), 597--614.

\bibitem[{\textit{Smythe-Wright}(1999)}]{Smythe-Wright:99}
Smythe-Wright (1999), {RRS} '{D}iscovery' cruise 233, 23 {A}pr-01 {J}un 1998.
  {A} {C}hemical and {H}ydrographic {A}tlantic {O}cean {S}urvey: {CHAOS},
  \textit{Southampton Oceanography Centre Cruise Report}, \textit{24}, 86pp.

\bibitem[{\textit{Tsuchiya et~al.}(1992)\textit{Tsuchiya, Talley, and
  Cartney}}]{Tsuchiya:92}
Tsuchiya, M., L.~D. Talley, and M.~S.~M. Cartney (1992), An eastern {A}tlantic
  section from {I}celand southward across the equator, \textit{Deep-Sea Res.},
  \textit{39}, 1185--1917.

\bibitem[{\textit{Wanninkhof}(1992)}]{Wanninkhof:92}
Wanninkhof, R. (1992), Relationship between wind speed and gas exchange over
  the ocean, \textit{J. Geophys. Res.}, \textit{97}, 7373--7382.

\end{thebibliography}


\end{article}

\end{document}
